

Engineering of Genetically Arrested Parasites (GAPs) For a Precision Malaria Vaccine

Oriana Kreutzfeld*, Katja Müller* and Kai Matuschewski*

Department of Molecular Parasitology, Institute of Biology, Humboldt University, Berlin, Germany

Continuous stage conversion and swift changes in the antigenic repertoire in response to acquired immunity are hallmarks of complex eukaryotic pathogens, including *Plasmodium* species, the causative agents of malaria. Efficient elimination of *Plasmodium* liver stages prior to blood infection is one of the most promising malaria vaccine strategies. Here, we describe different genetically arrested parasites (GAPs) that have been engineered in *Plasmodium berghei*, *P. yoelii* and *P. falciparum* and compare their vaccine potential. A better understanding of the immunological mechanisms of prime and boost by arrested sporozoites and experimental strategies to enhance vaccine efficacy by further engineering existing GAPs into a more immunogenic form hold promise for continuous improvements of GAP-based vaccines. A critical hurdle for vaccines that elicit long-lasting protection against malaria, such as GAPs, is safety and efficacy in vulnerable populations. Vaccine research should focus on solutions toward turning malaria into a vaccine-preventable disease, which would offer an exciting new path of malaria control.

OPEN ACCESS

Edited by:

Nathan W. Schmidt, University of Louisville, United States

Reviewed by:

Jason Scott Stumhofer, University of Arkansas for Medical Sciences, United States Scott E. Lindner, Pennsylvania State University, United States

*Correspondence:

Oriana Kreutzfeld kreutzfo@hu-berlin.de Katja Müller muekatja@hu-berlin.de Kai Matuschewski kai.matuschewski@hu-berlin.de

> Received: 14 March 2017 Accepted: 04 May 2017 Published: 31 May 2017

Citation:

Kreutzfeld O, Müller K and Matuschewski K (2017) Engineering of Genetically Arrested Parasites (GAPs) For a Precision Malaria Vaccine. Front. Cell. Infect. Microbiol. 7:198. doi: 10.3389/fcimb.2017.00198 Keywords: malaria, Plasmodium, vaccine, liver stage, live attenuated parasite, sporozoite, immune memory

INTRODUCTION

Malaria remains the most important vector-borne infectious disease and affects half of the world's population. Globally, >200 million infected individuals develop clinical symptoms, and >400,000 die because of severe malaria, primarily children in Sub-Saharan Africa (WHO, 2016a). New strategies for malaria prevention and eradication are thus urgently required. The current malaria control programs target the causative agents, *Plasmodium falciparum*, *P. vivax*, and three other human-infecting *Plasmodium* parasites, at different life cycle stages, which together reduces morbidity and mortality in endemic regions. Attacking the parasite at its vector stage with long-lasting insecticide treated bed nets and insecticide-indoor residual spraying combined with access to rapid diagnosis and artemisinin-based combination therapy for clinical malaria episodes are recommended by the WHO (WHO, 2016a).

Abbreviations: $\beta 2M$, β_2 microglobulin; CAS, chemically arrested sporozoites; CSP, circumsporozoite protein; DC, dendritic cell; ETRAMP, early transcribed membrane protein; FASII, fatty acid synthesis II; G3PAT, glycerol-3-phosphate acyltransferase; G3PDH, glycerol-3-phosphate dehydrogenase; GAP, genetically arrested parasite; IK2, eukaryotic initiation factor- 2α kinase; MHC, major histocompatibility complex; PALM, *Plasmodium*-specific apicoplast protein important for liver merozoite formation; PDH, pyruvate dehydrogenase; PKG, protein kinase G; PP1, eIF2 α -P protein phosphatase 1; MSP1, merozoite surface protein 1; PVM, parasitophorous vacuole membrane; RAS, radiation attenuated sporozoites; SLARP/SAP1, sporozoite and liver stage asparagine-rich protein; TRAP, thrombospondin-related anonymous protein; UIS, upregulated in infectious sporozoites.

Repeated exposure to Plasmodium transmission in malaria-endemic countries leads only to very slow acquisition of naturally acquired immunity that rapidly wanes. Development of protective immunity is likely hindered by blood infection, the exclusive cause of malaria-related morbidity and mortality. Therefore, the pre-erythrocytic phase of the Plasmodium life cycle in the mammalian host is particularly attractive as immunization agent since no clinical symptoms are associated with this first replication phase (Prudêncio et al., 2006; Silvie et al., 2008a; Matuschewski et al., 2011). It also allows priming of CD4⁺ and CD8⁺ T cells by presenting parasite antigens to the host immune system via MHCI and MHCII by dendritic cells and MHCI by infected hepatocytes, respectively (Hafalla et al., 2011). Prevention of disease by vaccination is an ideal addition to the portfolio of malaria intervention tools; however, it remains one of the greatest challenges in medical research.

Approaches to design a protective long-lasting malaria vaccine are wide-ranging, and some of them have already reached clinical trial phases (WHO, 2016b). Among these candidates is RTS,S/AS01, which is the first malaria vaccine that achieved licensure for administration in malaria endemic areas. RTS,S/AS01 is a hepatitis B-based subunit vaccine and contains a fragment of the circumsporozoite protein (CSP) (Stoute et al., 1997). Recent phase III clinical trial results revealed an unfavorable protection level casting doubt on the impact of RTS,S/AS01 in malaria control efforts (RTS,S Clinical Trials Partnership, 2011; Olotu et al., 2016).

A benchmark for a malaria vaccine exists for several decades already, and this experimental vaccine with proven long-term protection is a whole sporozoite vaccine. Small immunization studies in mice, non-human primates, and humans demonstrated that radiation attenuated sporozoites (RAS) elicit sterile protection against Plasmodium challenge infections (Nussenzweig et al., 1967, 1969; Clyde et al., 1973; Gwadz et al., 1979; Hoffman et al., 2002). Irradiation induces DNA breakage in the parasites, which reduces the numbers of nuclear divisions and limits liver stage expansion to early schizonts (Silvie et al., 2002). Timing and radiation dosage is critical since over-irradiated sporozoites arrest early at the unicellular stage, thus decreasing protection in experimental cohorts (Friesen and Matuschewski, 2011). Successful hepatocyte invasion and initial intra-hepatic development of live, metabolically active sporozoites are required to elicit strong immunity, and, therefore, pose a substantial hurdle toward an affordable and undemanding sporozoite vaccine. The central importance of live and metabolically active sporozoites has been corroborated with heat-killed sporozoites, which elicit only very weak and short-term, antibody-mediated protection against subsequent Plasmodium sporozoite challenge infections (Hafalla et al., 2006). On the other hand, suboptimal irradiation harbors the risk of breakthrough infections during vaccinationa considerable drawback concerning safety. RAS arrest early during liver stage development and express primarily sporozoitederived antigens, including CSP and thrombospondin-related anonymous protein (TRAP), which likely contribute to priming of T cell-mediated immunity.

Liver stage developmental arrest can also be achieved by an alternative approach, termed chemical attenuation of sporozoites (CAS), which is based on simultaneous administration of normal sporozoites and anti-malarial drugs (Belnou et al., 2004; Putrianti et al., 2009; Roestenberg et al., 2009; Friesen et al., 2010; Bijker et al., 2013). While they offer interesting evaluations in small-scale exploratory clinical studies, these approaches are critically reliant on continuous clinical supervision during drug administration and currently bear no translational perspective.

The potential to induce lasting protection by live attenuated, metabolically active parasites led to the engineering of genetically attenuated parasites (GAPs) as tailored whole parasite vaccines against malaria infections (Mueller et al., 2005a). Murine malaria models employing the rodent malaria parasites P. berghei and P. yoelii enable the exploration of liver stage-specific proteins and their importance for parasite survival. In the past years, over 120 genes have been targeted by experimental genetics and analyzed for defects during the Plasmodium life cycle in both vector and mammalian host (Janse et al., 2011). Many gene knockouts resulted in normal parasite life cycle progression or refractoriness to targeted deletion, indicative of redundant functions during life cycle progression or vital roles for blood infection, respectively. Additionally, arrest of the parasite development in the mosquito midgut, prior to salivary gland colonization, or ahead of hepatocyte invasion was frequently observed. Accordingly, only very few candidate genes fulfill the criteria of potential GAP vaccine candidate lines (Table 1).

Herein, we review the most recent developments in GAP vaccine discovery. We assess the different GAPs that have been generated in the last 12 years and evaluate their vaccine potential. Finally, we address the challenges and obstacles in designing a GAP vaccine for vulnerable populations in malaria-endemic countries.

FIRST GENERATION GAPs: PROOF OF PRINCIPLE STUDIES

The first preclinical studies that showed successful generation of GAP lines and their efficacy in immunization protocols targeted Plasmodium berghei genes that represent members of the early transcribed membrane protein (ETRAMP) family (Mueller et al., 2005a,b). The selected genes, i.e., upregulated in infective sporozoites gene 3 (UIS3) and UIS4, fulfilled three principal criteria: (i) stage-specific gene expression in pre-erythrocytic stages, thereby allowing recombinant knockout strains to be selected during blood stage transfection; (ii) abundant expression in pre-erythrocytic stages, indicating likely vital roles during this phase of the life cycle; and (iii) genes that are unique to Plasmodium and related haemosporidian parasites, which all share the hallmark of the first obligate population expansion phase in the host liver (Matuschewski et al., 2002). In vitro studies in cultured hepatoma cells revealed that P. berghei as well as P. yoelii $\triangle UIS3$ and $\triangle UIS4$ parasite lines arrest early in liver stage development after completion of sporozoite transformation to liver stages, but before onset of parasite

Target gene	Gene IDs	Liver stage arrest	Breakthroughs	References ^a	
ETRAMPs/PV PROTEI	NS				
UIS3	PBANKA_1400800	Early	No	Mueller et al., 2005a	
	PY17X_1402400	Early	No	Tarun et al., 2007	
UIS4	PBANKA_0501200	Early	Yes	Mueller et al., 2005b	
	PY17X_0502200	Early	No	Tarun et al., 2007	
6-CYS PROTEINS					
P36, P36p (P52)	PBANKA_1002100,	Early	Yes	van Dijk et al., 2005,	
	PBANKA_1002200	Early		Ishino et al., 2005	
	PY17X_1003500,	Early	No	Labaied et al., 2007	
	PY17X_1003600	Early			
	PF3D7_0404400,	Early	Yes	van Schaijk et al., 2008	
	PF3D7_0404500	Early			
B9	PBANKA_0808100	Early	Yes	Annoura et al., 2014	
	PF3D7_0317100	Early	Yes	van Schaijk et al., 2014a	
DIFFERENTIATION FA	CTORS				
SLARP (SAP1)	PBANKA_ 090210	Early	No	Silvie et al., 2008b	
	PY17X_0903500	Early	No	Aly et al., 2008	
	PF3D7_1147000	Early	No	van Schaijk et al., 2014a	
METABOLISM					
Fabl	PBANKA_1229800	Late	Yes	Yu et al., 2008	
FabB/F	PBANKA_1125100	Late	Yes	Annoura et al., 2012	
	PY17X_1126500	Late	No	Vaughan et al., 2009	
FabZ	PY17X_1342900	Late	No	Vaughan et al., 2009	
G3PAT	PBANKA_1416700	Late	Yes	Shears et al., 2017	
	PY17X_1418400	Late	No	Lindner et al., 2014	
G3PDH	PY17X_0934900	Late	No	Lindner et al., 2014	
PDH-E1a	PBANKA_0923800	Late	Yes	Nagel et al., 2013	
	PY17X_0925800	Late	No	Pei et al., 2010	
PDH-E3	PY17X_0715100	Late	No	Pei et al., 2010	
TRANSPORTERS					
MFS6	PBANKA_1304700	Late	Yes	Kenthirapalan et al., 2016	
FUNCTION UNKNOWN	4				
LISP2	PBANKA_1003000	Late	Yes	Orito et al., 2013	
PALM	PBANKA_0101100	Very late	Yes	Haussig et al., 2011	
PlasMei2	PY17X_1123700	Late	No	Dankwa et al., 2016	
SPELD	PBANKA_0910900	Mid	Yes	Al-Nihmi et al., 2017	
PKG ^b	PBANKA_1008200	Late	No	Falae et al., 2010	

TABLE 1 | Plasmodium genes targeted for GAPs. Plasmodium berghei (black), P. yoelii (green) and P. falciparum (blue) are listed.

^aReferences list only the first report in the respective Plasmodium species.

^bIncomplete stage-specific gene knockout.

replication (Figure 1; Mueller et al., 2005a,b; Tarun et al., 2007).

Immunization studies with one prime and two booster vaccinations showed complete long-term protection with *P. berghei* and *P. yoelii* $\Delta UIS3$ and $\Delta UIS4$ GAPs in C57BL/6 and BALB/cJ mice, respectively (**Table 2**) (Mueller et al., 2005a,b; Tarun et al., 2007). The study on *P. berghei* $\Delta UIS4$ GAPs established frequent occurrence of breakthrough infections, *i.e.*, a proportion of animals develop blood infections during the immunization procedure (Mueller et al., 2005b). This safety concern needs careful examination

before progression of *P. falciparum* GAPs to human testing can take place (Matuschewski, 2013). Notably, the identification of orthologous genes for *UIS3* and *UIS4* in *P. falciparum* has so far remained elusive, although various ETRAMPs emerge as potential candidates (Spielmann et al., 2012). To date, only one *P. falciparum ETRAMP* has been targeted by reverse genetics (MacKellar et al., 2010). This protein, termed *Pf*ETRAMP 10.3, apparently performs essential functions during blood infection, since it remains refractory to targeted deletion and does not complement the *P. yoelii* $\Delta UIS4$ defects. Accordingly, more research is needed to identify and target *P. falciparum* ETRAMPs



to generate this class of GAPs in *P. falciparum* and test their vaccine potential in small-scale human trials.

Another first generation GAP line in *P. berghei* and *P. yoelii* are parasites that lack *P36p* (also termed *P52*), a member of the *Plasmodium*-specific 6-Cys protein family (van Dijk et al., 2005). Parasites lacking *P36p* (*P52*) or its paralog, *P36*, arrest again early during liver stage development after completion of sporozoite transformation (**Figure 1**) (Ishino et al., 2005; van Dijk et al., 2005). Δ *P36p* sporozoites transmigrate and invade hepatocytes, wherein they initiate the formation of a parasitophorous vacuole membrane (PVM). Although, they start to mature into liver stage trophozoites, Δ *P36p* parasites suddenly abort this development, most likely because maintenance and maturation of the PVM are critically impaired (van Dijk et al., 2005).

Once more, occasional breakthrough infections in mice inoculated with $\Delta P36p$ sporozoites were observed (van Dijk et al.,

2005). Interestingly, in the $\Delta P36p/P36$ double knockout parasite expression of the signature merozoite surface protein 1 (MSP1) could not be detected, and the mechanism of breakthrough infections remains unsolved (Ploemen et al., 2012). Targeted gene deletion of a second member of the 6-Cys family, termed *B9*, led to similar early arrest and occasional breakthrough infections (Annoura et al., 2014).

Since genes of the 6-Cys protein family are remarkably conserved across *Plasmodium* species, *P. falciparum* GAP lines were generated to show proof of principle of liver stage attenuation by targeted gene deletion in human malarial parasites (van Schaijk et al., 2008; van Buskirk et al., 2009). However, in order to advance this approach to clinical testing in humans, safety is of utmost importance. Accordingly, it remains enigmatic why a *Pf* GAP based on *P52* (*P36p*) and *P36* was selected for a first phaseI/IIa clinical trial (Spring et al., 2013), despite the alarming

TABLE 2 | P. berghei and P. yoelii GAPs: long-term immunization studies.

GAPs ^a	Immunization dose (days after priming)		Challenge ^b (days after last boost)	No. protected/No. challenged animals ^c	References	
	1	2	3			
ETRAMPS						
∆UIS3	50 k	25 k (34)	25 k (45)	10 k (30)	5/5 (100%)	Mueller et al., 2005a
	10 k	10 k (34)	10 k (45)	10 k (30)	5/5 (100%)	
	10 k	10 k (14)	10 k (28)	10 k (60)	4/4 (100%)	Tarun et al., 2007
				(180)	8/12 (67%)	
	10 k	10 k (14)	_	10 k (30)	4/4 (100%)	
$\Delta UIS4$	50 k	25 k (14)	25 k (28)	50 k (38)	8/8 (100%)	Mueller et al., 2005b
	10 k	10 k (14)	10 k (28)	50 k (38)	8/8 (100%)	
	10 k	10 k (14)	10 k (28)	10 k (60)	4/4 (100%)	Tarun et al., 2007
			- (- /	(180)	8/8 (100%)	
	50 k	_	_	10 k (30)	4/4 (100%)	
AUIS3/UIS4	75 k	20 k (7)	20 k (14)	10 k (180)	6/6 (100%)	Jobe et al., 2007
20,000,010 /	10 k	10 k (14)	10 k (28)	10 k (118)	14/14 (100%)	0000 or all, 2001
AUIS3/LISP2	50 k	20 k (14)	20 k (28)	10 k (42)	8/8 (100%)	Kumar et al. 2016
	0011	2010(11)	2011(20)	(102)	8/8 (100%)	Ramar of all, 2010
6-CYS PROTEINS				(102)	0/0 (10070)	
ΔΡ36ρ	50 k	20 k (7)	20 k (14)	10 k (30)	5/5 (100%)	van Diik et al., 2005
AP52/P36	10 k	10 k (7)	10 k (14)	10 k (30)	7/7 (100%)	Labaied et al., 2007
Λ <i>B</i> 9	50 k	20 k (7)	20 k (14)	10 k (90)	5/5 (100%)	van Schaiik et al., 2014a
		()		(180)	9/9 (100%)	· · · · · · · · · · · · · · · · · · ·
				(365)	5/11 (45%)	
DIFFERENTIATION FACT	ORS			()		
$\Delta SLARP (\Delta SAP1)$	50 k	25 k (14)	25 k (28)	10 k (42)	5/5 (100%)	Silvie et al., 2008b
				(98)	2/5 (40%)	
	10 k	10 k (14)	10 k (28)	10 k (36)	1/4 (25%)	
	1 k	1 k (14)	1 k (28)	10 k (36)	0/4 (0%)	
	10 k	10 k (14)	10 k (28)	10 k (30)	25/25 (100%)	Aly et al., 2008
				10 k (120)	25/25 (100%)	
	1 k	1 k (14)	1 k (28)	10 k (30)	10/10 (100%)	Butler et al., 2011
	1 k	1 k (14)	_	10 k (30)	2/10 (20%)	
Δ SLARP/B9	50 k	25 k (7)	25 k (14)	10 k (180)	6/6 (100%)	van Schaiik et al., 2014a
∆SAP1/P52/P36	10 k	10 k (14)	_	10 k (30)	5/5 (100%)	Kublin et al., 2017
				(180)	5/5 (100%)	
FATTY ACID SYNTHESIS	PATHWAY				× 7	
$\Delta FabB/F$	50 k	50 k (14)	50 k (28)	10 k (210)	8/8 (100%)	Butler et al., 2011
	10 k	10 k (14)	10 k (28)	10 k (100)	5/5 (100%)	
				(300)	8/8 (100%)	
	10 k	10 k (14)	10 k (28)	10 k (30)	8/8 (100%)	
	1 k	1 k (14)	1 k (28)	10 k (30)	10/10 (100%)	
	1 k	1 k (14)	_	10 k (30)	10/10 (100%)	
$\Delta G3PAT$	10 k	10 k (37)	_	10 k (37)	30/30 (100%)	Lindner et al., 2014
$\Delta G3PDH$	10 k	10 k (37)	_	10 k (37)	30/30 (100%)	Lindner et al., 2014
$\Delta PDH-E1-PFO_{LS}$	5 k	5 k (25)	5 k (50)	5 k (30)	10/10 (100%)	Nagel et al., 2013
OTHER FUNCTIONS						-
ΔPALM	10 k	10 k (~30)	_	10 k (~30)	5/5 (100%)	Haussig et al., 2011
				(110)	6/7 (86%)	
$\Delta MFS6$	10 k	10 k (9)	_	10 k (22)	2/2 (100%)	Kenthirapalan et al., 2016
	1 k	1 k (9)	_	10 k (22)	1/12 (8%)	

^aP. berghei and P. yoelii knockout parasite lines are displayed in black and gray, respectively.

^bOnly studies where i. v. challenge was performed at least 3 weeks after the last boost are listed.

^cImmunizations were performed in the P. berghei-C57BL/6 and P. yoelii-BALB/cJ combinations, respectively.

preclinical data. As predicted, human trials with $Pf\Delta P52/P36$ had to be suspended because of breakthrough infections (Spring et al., 2013).

A potential, albeit untested, advantage of 6-Cys proteinbased GAPs is enhanced antigen presentation via MHCI on infected hepatocytes, since maintenance of the parasitophorous vacuole is impaired (van Dijk et al., 2005). Perhaps even more important, the propensity to increase apoptosis in $\Delta P36p$ infected hepatocytes (van Dijk et al., 2005) might enhance crosspriming by dendritic cells (DCs) that phagocytose apoptotic bodies (Leiriao et al., 2005), although this conjecture remains controversial (Renia et al., 2006).

Together, the first generation GAPs established that precise developmental arrests during the first clinically silent, intrahepatic *Plasmodium* expansion phase can be engineered by tailored removal of individual vital genes from the entire *Plasmodium* genome (**Table 1**). These uniform, genetically defined parasites consistently elicit lasting protection against sporozoite challenge infections in vaccination protocols with three consecutive GAP sporozoite inoculations (**Table 2**).

GAPs TARGETING LIVER STAGE DIFFERENTIATION: SAFETY FIRST

The observations of breakthrough blood infections during the immunization protocol in a proportion of animals (Mueller et al., 2005b; van Dijk et al., 2005) initiated the search for candidate Plasmodium genes that are key developmental factors, for instance transcription factors at the nexus of sporozoite to liver stage transformation. Such a factor was identified during the analysis of sporozoite-specific (S) genes (Kaiser et al., 2004). Targeted gene deletion of this factor, termed S22 or sporozoite and liver stage asparagine-rich protein (SLARP or SAP1) (Aly et al., 2008; Silvie et al., 2008b), resulted in a complete arrest of the parasite at early liver stage development prior to nuclear division (Figure 1). A novel hallmark of $\triangle SLARP$ parasites was the differential down-regulation of many liver stage-specific mRNAs and their corresponding proteins, including PVMresident proteins such as UIS3 and UIS4 (Aly et al., 2008; Silvie et al., 2008b), suggesting pleiotropic defects as a result of the absence of a major transcriptional regulator of liver stage differentiation.

Most importantly, Δ *SLARP* parasites are the first and only GAPs reported to fulfill all criteria of safe arrest (Aly et al., 2008; Silvie et al., 2008b). Accordingly, after the safety failure of Pf Δ *P52/P36*, all *Pf*GAPs developed for clinical testing in humans include the corresponding *PfSLARP* knockout (Mikolajczak et al., 2014; van Schaijk et al., 2014a; Kublin et al., 2017). However, immunization studies showed a lack in long-term protection in animals immunized with *P. berghei* Δ *SLARP* parasites, where only 40% of all animals were protected 3.5 months after immunization (**Table 2**) (Silvie et al., 2008b). It is conceivable that Δ *SLARP* parasites display a smaller array of antigens, but this has to be experimentally tested employing systems immunology approaches.

Notably, targeting of other key factors important for liver stage gene expression did not result in a similar complete liver stage arrest. Deletion of the liver stage-specific transcription factor of the apetala 2 family, termed *AP2-L*, only resulted in a developmental delay (Iwanaga et al., 2012). Knockout of the eukaryotic initiation factor 2α (eIF2 α) kinase (*IK2/UIS1*), which is critical for sporozoite latency, and stage-specific knockout of the corresponding eIF2 α -P protein phosphatase 1 (*PP1/UIS2*), which is a regulator of protein translation after hepatocyte invasion, led to incomplete early arrests before and immediately after hepatocyte invasion (Zhang et al., 2010, 2016). Targeted deletion of another regulator of sporozoite latency, the RNAbinding protein *PUF2*, reproduced the $\Delta IK2/UIS1$ phenotype, again with an unsatisfactory safety profile (Gomes-Santos et al., 2011; Müller et al., 2011).

In conclusion, immunization data together with the demonstration of a very early, complete arrest indicate that Δ *SLARP* parasites are comparable to RAS, with the important distinctions of a precision life cycle arrest in humans (Kublin et al., 2017) and safe handling of Δ *SLARP*-infected *Anopheles* mosquitoes for vaccine production.

LATE ARRESTING GAPs: IMPROVED IMMUNOGENICITY BUT LACK OF SAFETY

Studies employing co-administration of normal sporozoites and anti-malarial drugs have consistently shown superior immunity of late liver stage and/or early blood cycle arrest in murine malaria models (Belnou et al., 2004; Friesen et al., 2010; Friesen and Matuschewski, 2011) and small scale human trials (Roestenberg et al., 2009; Bijker et al., 2013), suggesting that a late liver stage arrest offers multiple advantages, perhaps including broader antigen presentation (Borrmann and Matuschewski, 2011). Unexpectedly, antibiotic-induced arrest at the transition from late liver stages to blood infection leads to better protection than a later arrest, after a few rounds of blood stage replication, induced by chloroquine treatment (Friesen and Matuschewski, 2011). This indicates an immune-modulatory effect by infected red blood cells and potential benefits of a complete arrest at the liver stage.

Therefore, tailored arrest toward the end of liver stage maturation was an important third step in GAP vaccine design. A CAS-based arrest using the antibiotic azithromycin showed that specific targeting of the *Plasmodium* apicoplast, a relict non-photosynthetic plastid organelle, resulted in late arrest after complete liver stage maturation (Friesen et al., 2010). Accordingly, two complementary approaches targeting key factors in the *Plasmodium* apicoplast led to generation of late-arrested GAPs and their testing in vaccine studies (**Figure 1**), namely deletion of a fatty acid biosynthesis enzyme (Butler et al., 2011) and a *Plasmodium*-specific protein of unknown function (Haussig et al., 2011). In both cases potent protection against reinfection was reported and superior protection correlated with extended liver stage maturation.

The fatty acid synthesis II (FASII) pathway produces saturated fatty acids in the apicoplast. It includes a cyclic reaction

that catalyzes fatty acid elongation and a large pyruvate dehydrogenase (PDH) complex that forms acetyl-CoA for the elongation cycles (Yu et al., 2008). Targeted deletion of an enzyme of the cyclic reaction, namely *trans*-2-enoyl-ACP reductase (*Fab1*), in *P. berghei* revealed a specific defect during liver stage maturation (Yu et al., 2008). However, the life cycle arrest was incomplete and sporozoite inoculations resulted in substantial breakthrough infections in C57BL/6 mice (**Table 1**) (Yu et al., 2008). Accordingly, immunization studies were not performed.

The liver stage defects of mutants in the FASII biosynthesis pathway in murine malaria parasites were confirmed in P. yoelii studies by targeted deletion of 3-oxoacyl-ACP synthase I/II (*FabB*/*F*) and β -hydroxyacyl-ACP dehydratase (*FabZ*) (Vaughan et al., 2009). Since breakthrough infections were once more absent in the *P. yoelii* model testing of $\Delta FabB/F$ parasites as latearrested GAPs in vaccination protocols was possible (Table 2) (Butler et al., 2011). This study reported better protection and a larger CD8⁺ T cell response in comparison to $\Delta SAP1$ and RAS parasites. Targeted gene deletion of additional P. yoelii enzymes of the FASII and the subsequent lipid biosynthesis pathway resulted in similar arrests in liver schizont maturation prior to merozoite formation (Pei et al., 2010; Lindner et al., 2014). The target genes were the E1 α and E3 subunits of PDH, glycerol-3-phosphate acyltransferase (G3PAT), and glycerol-3phosphate dehydrogenase (G3PDH) (Figure 1) of which the latter performed well in vaccine protocols (Table 2).

The previous observation of breakthrough infections in the first study of a *P. berghei* FASII pathway knockout (Yu et al., 2008) was subsequently confirmed (**Table 1**) (Annoura et al., 2012; Shears et al., 2017) and strictly limits the results from studies in *P. yoelii*. The additional, unexpected finding of aborted parasite development in mosquitos infected with the corresponding *P. falciparum* knockouts (Cobbold et al., 2013; van Schaijk et al., 2014b) essentially eliminated the possibility to develop GAP vaccines by targeted deletion of the FASII pathway. Together, these results also raise the important question, which prerequisites have to be fulfilled to transfer discoveries from murine models to *Pf*GAP vaccines.

The second strategy of targeting essential apicoplast functions to generate late liver stage-arrested parasite lines built upon bioinformatic prediction of Plasmodium-specific apicoplast targeted proteins (Haussig et al., 2011). The first target that satisfied these criteria was P. berghei Plasmodium-specific apicoplast protein important for liver merozoite formation (PALM), a protein of unknown function. Targeted deletion of PALM did not affect parasite growth or apicoplast morphology, but resulted in an even later arrest after completion of liver stage development prior to merozoite release (Haussig et al., 2011). Although, immunizations with $\Delta PALM$ resulted in robust long-term protection after only two immunizations in the stringent P. berghei-C57BL/6 model (Table 2), consistent doseindependent breakthrough infections preclude the translation to human Plasmodium species, unless this mutant is combined with an additional knockout that causes a similar life cycle arrest. Several attempts to target other biochemical pathways within the apicoplast, including iron-sulfur cluster biogenesis or primary reactions in heme biosynthesis, did not yield $\Delta PALM$ -like GAPs (Haussig et al., 2013, 2014; Rizopoulos et al., 2016).

Additional late arresting candidate genes have been identified; however, in depth analysis and immunization studies are often lacking. An interesting case is protein kinase G (PKG), which is shared across different life cycle stages and already essential during blood infection, as it plays prominent roles in merozoite egress and gametogenesis. Since generation of a ΔPKG parasite is incompatible with blood infection, a stagespecific knockout by FLP/FRT recombination in sporozoites was engineered to study the role(s) in pre-erythryocytic development (Falae et al., 2010). This analysis revealed a late arrest in liver stage maturation and wild type breakthrough infections due to incomplete recombination (Table 1). Immunization studies were not conclusive since challenge infections were performed only one week after high dose immunizations. However, the example of PKG illustrates that a tight late arrested parasite might be achieved once a suitable liver stage-specific gene at the nexus of stage conversion is identified.

GAPs WITH MULTIPLE GENE DELETIONS: SYNERGISTIC OR ANTAGONISTIC?

Development of parasite lines that harbor multiple gene deletions is likely to increase safety; however, whether synergistic or antagonistic effects modulate immunogenicity is less straightforward to predict and likely depends on the selected knockout combination.

Soon after the first proof-of-principle studies a GAP parasite line that harbors two consecutive gene deletions, namely $\Delta UIS3$ and $\Delta UIS4$, was engineered in *P. berghei* (Jobe et al., 2007). As expected, $\Delta UIS3/UIS4$ parasites displayed a complete arrest in early liver stages, indicating that vaccine strains harboring multiple independent gene deletions perform safer than single knockout GAPs. Importantly, long-term protection against a high-dose sporozoite challenge infection was complete (**Table 2**).

In marked contrast, when a double knockout was performed for the P. berghei paralogs P36p (P52) and P36, which are neighboring genes and likely arose through gene duplication, safety was not improved (Annoura et al., 2012), strongly suggesting that independent genes need to be targeted in multiple gene knockout strategies. Since \triangle SLARP GAPs lead to a complete termination of liver stage development (Aly et al., 2008; Silvie et al., 2008b), they constituted the obvious platform for combinations with knockouts of the 6-Cys gene family (Figure 1) (Mikolajczak et al., 2014; van Schaijk et al., 2014a; Kublin et al., 2017). However, it remains to be shown whether addition of B9, P36, and/or P36p (P52) knockout provides any additional benefit beyond perception of additional gene deletions. An important investigation with combinatorial knockouts will be the systematic expression profiling of liver stage-specific genes, as was previously done for \triangle *SLARP* parasites (Silvie et al., 2008b). Such an analysis will provide first insights into the expected antigenic repertoire displayed by the respective GAPs.

Instead of adding additional gene deletions, which might not add significantly to vaccine efficacy, combination of a gene deletion with transgene expression of additional factors could amplify antigen presentation during pre-erythrocytic development. Potential avenues include perforation of the intracellular niche, i.e., the parasitophorous vacuole, activation of innate immune sensing pathways, and expression of blood stage and gametocyte antigens. Such a strategy is exemplified by the expression of perfringolysin O, a cholesterol-dependent cytolysin, in *P. berghei* ΔPDH -E1 α parasites (Nagel et al., 2013). Addition of the transgene could substantially reduce, albeit not completely abolish, breakthrough infections of the single gene deletion, providing a rationale for further bioengineering efforts to achieve premature rupture of the PVM. However, a principal concern in gain-of-function mutants is that parasites containing mutations in the transgene promoter and other regulatory elements, which lead to a reduced transgene expression, will be swiftly selected. Therefore, GAPs that express additional Plasmodium antigens in order to broaden the immunogenic repertoire without further reducing parasite fitness might be a particularly rewarding research direction.

IMMUNE MECHANISMS OF GAPs: THE CENTRAL ROLE OF EFFECTOR MEMORY CD8⁺ T CELLS

Cytolytic, interferon gamma (IFN γ)-secreting CD8⁺ T cells were identified early on as the key mediators of protection after radiation-attenuated sporozoite immunizations (Schofield et al., 1987; Weiss et al., 1988; Romero et al., 1989). Amongst all CD8⁺ T cell subsets, presence of those with an effector memory phenotype, i.e., CD45RBlo, CD44high, CD62Llo, and CD122^{lo} consistently correlated with long-lasting protection (Guebre-Xabier et al., 1999; Berenzon et al., 2003). Studies employing immunization of β_2 microglobulin ($\beta 2M$) knockout mice and recognition of a CSP-specific T cell clone by hepatocytes with a matching MHCI haplotype revealed that liver stage antigens are presented to the CD8⁺ T cells via MHCI on the surface of infected hepatocytes (White et al., 1996; Balam et al., 2012). Elimination of infected hepatocytes in immunized mice could be linked to perforin secreting cytotoxic CD8⁺ T cells and IFNy production (Mellouk et al., 1987; Schofield et al., 1987; Malik et al., 1991; Rodrigues et al., 1993; Sano et al., 2001). Employing advanced in vivo imaging techniques direct proximity to antigen-specific CD8⁺ T cells was shown to be required for cytolytic killing of Plasmodium-infected hepatocytes (Cockburn et al., 2013; Kimura et al., 2013). Although, antigenspecific CD4⁺ T cells are expected to be central to mount an effective T cell response, their roles in B cell help to produce antibodies that inhibit sporozoite attachment and invasion of the liver is only minor. Antibodies are not essential for vaccineinduced protection, and protective CD8⁺ T cell responses can be mounted without CD4⁺ help (Schofield et al., 1987; Rodrigues et al., 1993).

Based on the insights from studies with irradiated sporozoites, $CD8^+$ T cell-dependent elimination most likely is the immune effector mechanism in GAP-immunized animals. Indeed, a study employing *P. berghei* $\Delta UIS3$ parasites showed that

immunizations of B and T cell-deficient $rag1^{-/-}$ and $IFN\gamma^{-/-}$ knockout mice failed to induce protection, confirming the central role of adaptive immune responses leading to IFN γ production in vaccine-induced immunity (Mueller et al., 2007). Immunizations of B cell-deficient mice and common laboratory mice after depletion or adoptive transfer of CD4⁺ and CD8⁺ T cells fully corroborated the central importance of CD8⁺ T cells, but neither of antibodies nor of CD4⁺ T cells. Of note, primaquine treatment results in efficient cure of liver stage parasites and reversed $\Delta UIS3$ -mediated protection (Mueller et al., 2007). This is in perfect agreement with the requirement for parasite persistence as metabolically active, cell cycle arrested intra-hepatic stages in order to maintain long-term protection (Scheller and Azad, 1995).

Another early study demonstrated the central role of CD8⁺ T cells, and particularly IFN γ -secreting effector memory cells, in protection induced by a *P. berghei* $\Delta UIS3/UIS4$ double knockout parasite line (Jobe et al., 2007). Immunization of $\beta 2m^{-/-}$ mice, which are deficient in MHCI expression and CD8⁺ T cells, abrogated vaccine efficacy. The direct comparison to RAS showed that both immunization strategies induce similar immune responses, but GAP-immunized animals displayed consistently higher levels of IFN γ -secreting effector memory cells (Jobe et al., 2007). Accordingly, it is conceivable that the insights gained from RAS immunizations can be extrapolated at least to $\Delta UIS3/UIS4$ GAP vaccines.

The findings obtained with P. berghei GAPs were fully supported by a study reporting that depletion of CD8⁺ T cells, but neither CD4⁺ T cells nor IgG1 antibodies, abolished protection by P. yoelii $\Delta UIS3$ or $\Delta UIS4$ parasite lines (Tarun et al., 2007). Notably, CD8⁺ T cells from P. yoelii GAP immunized mice induce apoptosis of infected, in vitro cultured hepatocytes by contact dependent, perforinmediated cytotoxic killing, with only partial involvement of IFNy (Trimnell et al., 2009). Short-lived CD11a^{hi}, CD62L^{lo}, CD44^{hi} antigen-experienced effector CD8b⁺ T cells, which also express CD11c, expanded swiftly after one immunization with P. yoelii $\Delta UIS4$ sporozoites (Cooney et al., 2013). Furthermore, these cells are KLRG1⁺CD127⁻ terminal effector cells, which upon restimulation with infected hepatocytes secrete IFNy, TNF, and IL-2 and express CD107a and perforin. Late-arresting P. yoelii GAPs induced larger CD8⁺ T cell responses in comparison to RAS, and the CD8α^{lo}CD11a^{hi} effector memory phenotype, characterized by low expression levels of CD27, CD62L, and CD127, was increased; however, this study was done in C57BL/6 mice only (Butler et al., 2011).

While the cellular mechanisms that lead to elimination of infected hepatocytes after immunization are relatively well understood, the target epitopes displayed by infected hepatocytes remain less clear (Hafalla et al., 2011). There is growing evidence that a combination of epitopes rather than a single protective antigen correlates with vaccine-induced protection (Grüner et al., 2007; Hafalla et al., 2013). The present list of pre-erythrocytic T cell epitopes in H2-K^d-restricted BALB/cJ and H2-K^b-restricted C57BL/6 mice remains short (Romero et al., 1989; Hafalla et al., 2013; Murphy et al., 2013; Lau et al., 2014; Müller et al., 2017). The few immunogenic epitopes identified thus far underscore the fundamental differences of immunity elicited by eukaryotic pathogens in comparison to viruses and bacteria (Hafalla et al., 2013).

The mechanisms leading to efficient priming of effector immune responses against pre-erythrocytic parasites are still incompletely understood. Efficient priming almost certainly requires cross-presentation by dendritic cells (DCs) and is most likely a combination of DC-mediated antigen presentation in different organs, including skin-draining lymph nodes, spleen, and liver (Sano et al., 2001; Chakravarty et al., 2007; Cockburn et al., 2011; Balam et al., 2012). Most studies focused on the CSP epitope, which is abundantly expressed on the sporozoite surface and only recognized in H2-K^d -restricted BALB/cJ mice (Romero et al., 1989; Sano et al., 2001; Chakravarty et al., 2007; Balam et al., 2012). Whether GAPs aborting development at the trophozoite stage or later present additional antigens explaining superior CD8⁺ T cell responses remains elusive.

TOWARD UNIFIED GUIDELINES FOR TRANSLATIONAL GAP RESEARCH

There are currently no rules established for GAP development in P. falciparum; however, defining a set of requirements that need to be fulfilled before moving on to P. falciparum studies would significantly decrease odds of clinical trial failures. An obvious criterion from murine malaria models, i.e., P. berghei and P. yoelii, is complete liver stage arrest in a large number of animals employing both murine models simultaneously. Humanized mouse models and primary human hepatocytes can provide preliminary indications on developmental arrest in the liver and, thus, potential safety concerns of a P. falciparum GAP vaccine candidate. However, these test systems might not provide the level of stringency necessary to predict breakthrough infections, which are presently best captured in preclinical tests in large groups of mice. In order to develop potent P. falciparum GAPs beyond early arresting $\Delta SLARP$ mutants, multiple independent gene deletions, ideally in unrelated physiological processes and in different cellular compartments, need to be considered.

Another desirable standard is robust long-term immunity against challenge infection. Identifying the correct parasite/host combination is important to draw valid conclusions from immunization experiments (Matuschewski, 2013). The P. berghei-C57BL/6 combination remains the most robust vaccine model to date, whereas immunizations using P. yoelii mutants in BALB/c mice are only of modest predictive value, since protection is easily achieved in this model. Other combinations, such as P. berghei and BALB/c mice, are invalid because of refractoriness of particular combinations of mouse strains and infections with certain murine malaria sporozoites (Scheller et al., 1994). This notion is illustrated in immunizations with *P. berghei* $\Delta P36p$ sporozoites, where only one immunization dose in an inappropriate host strain, BALB/c mice, resulted in complete protection up to 3 months after immunization (van Dijk et al., 2005). Immunization with P. yoelii $\Delta P36p/P36$ parasites also induced sterile protection

in BALB/c mice after a single injection (Labaied et al., 2007); however, such protection in C57BL/6 mice was only elicited after three rounds of immunization (van Dijk et al., 2005). One important immunological difference is that protection in the P. yoelii-BALB/c model largely depends on the immunedominant CD8⁺ T cell epitope of CSP. In marked contrast, T cell responses in the P. berghei-C57BL/6 model are likely multifactorial (Hafalla et al., 2011, 2013), which closely mimics infections in human populations with a large range of MHCI haplotypes and only infrequent CSP-specific CD8⁺ T cell responses (Offeddu et al., 2012). The recent evaluation of Grammomys dolichurus, an Afrotropical arboreal rodent, which naturally harbors rodent Plasmodium infections, as a model to study pre-erythrocytic vaccine strategies will be an important addition for preclinical evaluation of safety and immunogenicity of GAP vaccines (Conteh et al., 2017). In good agreement with the data from murine *Plasmodium* models, the natural host is highly susceptible to P. berghei sporozoite induced infections and multiple high immunization doses are required for robust protection (Conteh et al., 2017).

Together, comparative studies on vaccine efficacy should include the *P. berghei*-C57BL/6 model (Friesen and Matuschewski, 2011), to better define long-term protection against re-challenges. We propose that a GAP vaccine line should be evaluated for vaccine safety and immunogenicity in a two-step preclinical process before advancing to human trials. First, comparative evaluation of the candidate *P. berghei* and *P. yoelii* GAP lines in the respective mouse strains, C57BL/6 and BALB/c, will pre-select GAPs that are completely arrested upon high dose sporozoite inoculations and elicit long-lasting (>30 days) sterile immunity. Second, confirmation of safety and immunogenicity of the *P. berghei* GAP line in *G. dolichurus* immunization and challenge study provide an evidence-based rationale for translation to *P. falciparum* GAP trials.

ROADBLOCKS TOWARD TRANSLATION OF GAPs

Intravenous injections are the preferential route of vaccine administration in mouse models; however, in real life this is inapplicable. Pediatric vaccines are exclusively administered either by intramuscular syringe injection or orally. It is evident that a malaria vaccine must adhere to the same safe routes of administration. Alternative methods that are used in drug delivery, such as intradermal, intravenous, or intraperitoneal injections, are unreasonable for a malaria vaccine to be delivered in resource-poor health infrastructures, because of the risks associated with these routes of administration, except under physician's care.

Plasmodium sporozoites apparently lack the ability to transmigrate through muscle or fat tissue. Accordingly, after intradermal, intramuscular, subcutaneous, or intraperitoneal injection only a very small proportion of sporozoites reach a blood vessel, resulting in reduced liver infection. This was confirmed in human volunteer studies, where intramuscular or intradermal syringe injections of cryopreserved sporozoites

showed substantial delays in blood infections and high doses were required to consistently induce blood infections (Shekalaghe et al., 2014; Gómez-Pérez et al., 2015). As expected, protection after immunizations *via* these routes was negligible both in the *P. berghei* model and in volunteer studies with *P. falciparum* sporozoites (Epstein et al., 2011; Nganou-Makamdop et al., 2012). Thus, development of a GAP vaccine that can be administered in the muscle is a principal hurdle that needs to be overcome by bioengineering efforts.

The necessity for booster immunizations remains another critical limitation, which is unlikely to be removed. However, a reduction of the vaccine doses would facilitate the distribution of a malaria vaccine considerably. A better understanding of how to best amplify an initial protective $CD8^+$ T cell response toward a sustained effector-memory T cell response will be essential. It is conceivable that timing of antigen expression is important, and the temporal dynamics of expansion and contraction of antigen-specific T cells need to be analyzed to inform vaccine protocols (Hafalla et al., 2013; Murphy et al., 2013; Billman et al., 2016).

Cryopreservation of sporozoites is so far the only way of efficiently preserving the infectivity of attenuated sporozoites, posing huge logistic constraints. Thus, development of a vaccine formulation, which is stable under cooled conditions or, ideally, at ambient temperature and induces long-term protection, remains a critical bioengineering milestone. *Plasmodium* sporozoites are particularly sensitive to environmental conditions, and it remains entirely speculative whether an appropriate preservation process can be implemented. A GAP vaccine formulation might be further improved by addition of an adjuvant; however, no examples exist yet for live attenuated *Plasmodium* vaccines.

REFERENCES

- Al-Nihmi, F. M., Kolli, S. K., Reddy, S. R., Mastan, B. S., Togiri, J., Maruthi, M., et al. (2017). A novel and conserved Plasmodium sporozoite membrane protein SPELD is required for maturation of exo-erythrocytic forms. *Sci. Rep.* 7:40407. doi: 10.1038/srep40407
- Aly, A. S. I., Mikolajczak, S. A., Silva Rivera, H., Camargo, N., Jacobs-Lorena, V., Labaied, M., et al. (2008). Targeted deletion of SAP1 abolishes the expression of infectivity factors necessary for successful malaria parasite liver infection. *Mol. Microbiol.* 69, 152–163. doi: 10.1111/j.1365-2958.2008.06271.x
- Annoura, T., Ploemen, I. H., van Schaijk, B. C., Sajid, M., Vos, M. W., van Gemert, G. J., et al. (2012). Assessing the adequacy of attenuation of genetically modified malaria parasite vaccine candidates. *Vaccine* 30, 2662–2670. doi: 10.1016/j.vaccine.2012.02.010
- Annoura, T., van Schaijk, B. C., Ploemen, I. H., Sajid, M., Lin, J. W., Vos, M. W., et al. (2014). Two Plasmodium 6-Cys family-related proteins have distinct and critical roles in liver-stage development. *FASEB J.* 28, 2158–2170. doi: 10.1096/fj.13-241570
- Balam, S., Romero, J. F., Bongfen, S. E., Guillaume, P., and Corradin, G. (2012). CSP – a model for *in vivo* presentation of *Plasmodium berghei* sporozoite antigens by hepatocytes. *PLoS ONE* 7:e51875. doi: 10.1371/journal.pone. 0051875
- Belnou, E., Costa, F. T. M., Frankenberg, T., Vigario, A. M., Voza, T., Leroy, N., et al. (2004). Protective T cell immunity against malaria liver stage after vaccination with live sporozoites under chloroquine treatment. *J. Immunol.* 172, 2487–2249. doi: 10.4049/jimmunol.172.4.2487

OUTLOOK

Building on the success of live attenuated, metabolically active RAS, genetic engineering of liver stage-arrested parasites offers unprecedented opportunities to develop a precision malaria vaccine. Comparative studies with GAPs that display distinct temporal arrests during liver stage maturation provide a foundation for systems immunology approaches, which might in turn lead to a better mechanistic understanding of immune effector mechanisms that contribute to lasting protection against re-infection. Ultimately, evidence-based design of safe and effective whole sporozoite P. falciparum and P. vivax vaccines involves major research investments in preclinical murine malaria models before translation to the human parasite is warranted. Proper design of human clinical trials with predictive power for vaccine safety, negligible adverse events, and vaccine efficacy in young children living in very diverse tropical countries remains very challenging.

AUTHOR CONTRIBUTIONS

All authors contributed equally to this work and approved the manuscript for publication.

ACKNOWLEDGMENTS

The authors thank Diane Schad for the artwork on **Figure 1**. Work on genetically arrested parasites in murine malaria models is funded by the Deutsche Forschungsgemeinschaft through the graduate program 2046 "From experimental models to natural systems" (project B1).

- Berenzon, D., Schwenk, R. J., Letellier, L., Guebre-Xabier, M., Williams, J., and Krzych, U. (2003). Protracted protection to *Plasmodium berghei* malaria is linked to functionally and phenotypically heterogenous liver memory CD8+ T cells. *J. Immunol.* 171, 2024–2034. doi: 10.4049/jimmunol.171.4.2024
- Bijker, E. M., Bastiaens, G. J. H., Teirlinck, A. C., van Germert, G.-J., Graumans, W., van de Vegte-Bolmer, M., et al. (2013). Protection against malaria after immunization by chloroquine prophylaxis and sporozoites is mediated by preerythrocytic immunity. *Proc. Natl. Acad. Sci. U.S.A.* 110, 7862–7867. doi: 10.1073/pnas.1220360110
- Billman, Z. P., Kas, A., Stone, B. C., and Murphy, S. C. (2016). Defining rules of CD8(+) T cell expansion against pre-erythrocytic *Plasmodium* antigens in sporozoite-immunized mice. *Malar. J.* 15:238. doi: 10.1186/s12936-016-1295-5
- Borrmann, S., and Matuschewski, K. (2011). Targeting *Plasmodium* liver stages: better late than never. *Trends Mol. Med.* 17, 527–536. doi: 10.1016/j.molmed.2011.05.008
- Butler, N. S., Schmidt, N. W., Vaughan, A. M., Aly, A. S., Kappe, S. H., and Harty, J. T. (2011). Superior antimalarial immunity after vaccination with late liver stage-arresting genetically attenuated parasites. *Cell Host Microbe* 9, 451–462. doi: 10.1016/j.chom.2011.05.008
- Chakravarty, S., Cockburn, I. A., Kuk, S., Overstreet, M. G., Sacci, J. B., and Zavala, F. (2007). CD8+ T lymphocytes protective against malaria liver stages are primed in skin-draining lymph nodes. *Nat. Med.* 13, 1035–1041. doi: 10.1038/nm1628
- Clyde, D. F., Most, H., McCarthy, V. C., and Vanderberg, J. P. (1973). Immunization of man against sporozoite-induced falciparum malaria. Am. J. Med. Sci. 266, 169–177. doi: 10.1097/00000441-197309000-00002

- Cobbold, S. A., Vaughan, A. M., Lewis, I. A., Painter, H. J., Camargo, N., Perlman, D. H., et al. (2013). Kinetic flux profiling elucidates two independent acetyl-CoA biosynthetic pathways in *Plasmodium falciparum. J. Biol. Chem.* 288, 36338–36350. doi: 10.1074/jbc.M113.503557
- Cockburn, I. A., Amino, R., Kelemen, R. K., Kuo, S. C., Tse, S. W., Radtke, A., et al. (2013). *In vivo* imaging of CD8+ T cell-mediated elimination of malaria liver stages. *Proc. Natl. Acad. Sci. U.S.A.* 110, 9090–9095. doi: 10.1073/pnas.1303858110
- Cockburn, I. A., Tse, S.-W., Radtke, A. J., Srinivasan, P., Chen, Y.-C., Sinnis, P., et al. (2011). Dendritic cells and hepatocytes use distinct pathways to process protective antigens from *Plasmodium in vivo*. *PLoS Pathog*. 7:e1001318. doi: 10.1371/journal.ppat.1001318
- Conteh, S., Anderson, C., Lambert, L., Orr-Gonzalez, S., Herrod, J., Robbins, Y. L., et al. (2017). Grammomys surdaster, the natural host for *Plasmodium berghei* parasites, as a model to study whole-organism vaccines against malaria. *Am. J. Trop. Med. Hyg.* 96, 835–841. doi: 10.4269/ajtmh.16-0745
- Cooney, L. A., Gupta, M., Thomas, S., Mikolajczak, S., Choi, K. Y., Gibson, C., et al. (2013). Short-lived effector CD8 T cells induced by genetically attenuated malaria parasite vaccination express CD11c. *Infect. Immun.* 81, 4171–4181. doi: 10.1128/IAI.00871-13
- Dankwa, D. A., Davis, M. J., Kappe, S. H., and Vaughan, A. M. (2016). A *Plasmodium yoelii* Mei2-like RNA binding protein is essential for completion of liver stage schizogony. *Infect. Immun.* 84, 1336–1345. doi: 10.1128/IAI.01417-15
- Epstein, J. E., Tewari, K., Lyke, K. E., Sim, B. K., Billingsley, P. F., Laurens, M. B., et al. (2011). Live attenuated malaria vaccine designed to protect through hepatic CD8+ T cell immunity. *Science* 334, 475–480. doi: 10.1126/science.1211548
- Falae, A., Combe, A., Amaladoss, A., Carvalho, T., Ménard, R., and Bhanot, P. (2010). Role of *Plasmodium berghei* cGMP-dependent protein kinase in late liver stage development. *J. Biol. Chem.* 285, 3282–3288. doi: 10.1074/jbc.M109.070367
- Friesen, J., and Matuschewski, K. (2011). Comparative efficacy of pre-erythrocytic whole organism vaccine strategies against the malaria parasite. *Vaccine* 29, 7002–7008. doi: 10.1016/j.vaccine.2011.07.034
- Friesen, J., Silvie, O., Putrianti, E. D., Hafalla, J. C. R., Matuschewski, K., and Borrmann, S. (2010). Natural immunization against malaria: causal prophylaxis with antibiotics. *Sci. Transl. Med.* 2:40ra49. doi: 10.1126/scitranslmed.3001058
- Gomes-Santos, C. S., Braks, J., Prudêncio, M., Carret, C., Gomes, A. R., Pain, A., et al. (2011). Transition of Plasmodium sporozoites into liver stage-like forms is regulated by the RNA binding protein Pumilio. *PLoS Pathog.* 7:e1002046. doi: 10.1371/journal.ppat.1002046
- Gómez-Pérez, G. P., Legarda, A., Muñoz, J., Sim, B. K., Ballester, M. R., Dobaño, C., et al. (2015). Controlled human malaria infection by intramuscular and direct venous inoculation of cryopreserved *Plasmodium falciparum* sporozoites in malaria-naive volunteers: effect of infection volume and dose of infectivity rates. *Malar. J.* 14:306. doi: 10.1186/s12936-015-0817-x
- Grüner, A. C., Mauduit, M., Tewari, R., Romero, J. F., Depinay, N., Kayibanda, M., et al. (2007). Sterile protection against malaria is independent of immune responses to the circumsporozoite protein. *PLoS ONE* 2:e1371. doi: 10.1371/journal.pone.0001371
- Guebre-Xabier, M., Schwenk, R., and Krzych, U. (1999). Memory phenotype CD8(+) T cells persist in livers of mice protected against malaria by immunization with attenuated *Plasmodium berghei* sporozoites. *Eur. J. Immunol.* 29, 3978–3986. doi: 10.1002/(sici)1521-4141(199912)29:12<3978:: aid-immu3978>3.3.co;2-s
- Gwadz, R. W., Cochrane, A. H., Nussenzweig, V., and Nussenzweig, R. S. (1979). Preliminary studies on vaccination of rhesus monkeys with irradiated sporozoites of *Plasmodium knowlesi* and characterization of surface antigens of these parasites. *Bull. World Health Organ.* 57, 165–173.
- Hafalla, J. C., Bauza, K., Friesen, J., Gonzalez-Aseguinolaza, G., Hill, A. V., and Matuschewski, K. (2013). Identification of targets of CD8(+) T cell responses to malaria liver stages by genome-wide epitope profiling. *PLoS Pathog.* 9:e1003303. doi: 10.1371/journal.ppat.1003303
- Hafalla, J. C., Rai, U., Morrot, A., Bernal-Rubio, D., Zavala, F., and Rodriguez, A. (2006). Priming of CD8+ T cell responses following immunization with heat-killed *Plasmodium* sporozoites. *Eur. J. Immunol.* 36, 1179–1186. doi: 10.1002/eji.200535712

- Hafalla, J. C., Silvie, O., and Matuschewski, K. (2011). Cell biology and immunology of malaria. *Immun. Rev.* 240, 297–316. doi: 10.1111/j.1600-065X.2010.00988.x
- Haussig, J. M., Matuschewski, K., and Kooij, T. W. (2011). Inactivation of a Plasmodium apicoplast protein attenuates formation of liver merozoites. *Mol. Microbiol.* 81, 1511–1525. doi: 10.1111/j.1365-2958.2011.07787.x
- Haussig, J. M., Matuschewski, K., and Kooij, T. W. A. (2013). Experimental genetics of *Plasmodium berghei* NFU in the apicoplast iron-sulfur cluster biogenesis pathway. *PLoS ONE* 8:e67269. doi: 10.1371/journal.pone.0067269
- Haussig, J. M., Matuschewski, K., and Kooij, T. W. A. (2014). Identification of vital and dispensable sulphur utilization factors in the *Plasmodium* apicoplast. *PLoS ONE* 9:e89718. doi: 10.1371/journal.pone.0089718
- Hoffman, S. L., Goh, L. M., Luke, T. C., Schneider, I., Le, T. P., Doolan, D. L., et al. (2002). Protection of humans against malaria by immunization with radiationattenuated *Plasmodium falciparum* sporozoites. *J. Infect. Dis.* 185, 1155–1164. doi: 10.1086/339409
- Ishino, T., Chinzei, Y., and Yuda, M. (2005). Two proteins with 6-cys motifs are required for malarial parasites to commit to infection of the hepatocyte. *Mol. Microbiol.* 58, 1264–1275. doi: 10.1111/j.1365-2958.2005.04801.x
- Iwanaga, S., Kaneko, I., Kato, T., and Yuda, M. (2012). Identification of an AP2family protein that is critical for malaria liver stage development. *PLoS ONE* 7:e47557. doi: 10.1371/journal.pone.0047557
- Janse, C. J., Kroeze, H., van Wigcheren, A., Mededovic, S., Fonager, J., Franke-Fayard, B., et al. (2011). A genotype and phenotype database of genetically modified malaria-parasites. *Trends Parasitol.* 27, 31–39. doi: 10.1016/j.pt.2010.06.016
- Jobe, O., Lumsden, J., Mueller, A. K., Williams, J., Silva-Rivera, H., Kappe, S. H., et al. (2007). Genetically attenuated *Plasmodium berghei* liver stages induce sterile protracted protection that is mediated by major histocompatibility complex Class I-dependent interferon-gamma-producing CD8+ T cells. *J. Infect. Dis.* 196, 599–607. doi: 10.1086/519743
- Kaiser, K., Matuschewski, K., Camargo, N., Ross, J., and Kappe, S. H. I. (2004). Differential transcriptome profiling identifies *Plasmodium* genes encoding pre-erythrocytic stage-specific proteins. *Mol. Microbiol.* 51, 1221–1232. doi: 10.1046/j.1365-2958.2003.03909.x
- Kenthirapalan, S., Waters, A. P., Matuschewski, K., and Kooij, T. W. (2016). Functional profiles of orphan membrane transporters in the life cycle of the malaria parasite. *Nat. Commun.* 7:10519. doi: 10.1038/ncomms10519
- Kimura, K., Kimura, D., Matsushima, Y., Miyakoda, M., Honma, K., Yuda, M., et al. (2013). CD8+ T cells specific for a malaria cytoplasmic antigen form clusters around infected hepatocytes and are protective at the liver stage of infection. *Infect. Immun.* 81, 3825–3834. doi: 10.1128/iai.00570-13
- Kublin, J. G., Mikolajczak, S. A., Sack, B. K., Fishbaugher, M. E., Seilie, A., Shelton, L., et al. (2017). Complete attenuation of genetically engineered *Plasmodium falciparum* sporozoites in human subjects. *Sci. Transl. Med.* 9:eaad9099. doi: 10.1126/scitranslmed.aad9099
- Kumar, H., Sattler, J. M., Singer, M., Heiss, K., Reinig, M., Hammerschmidt-Kamper, C., et al. (2016). Protective efficacy and safety of liver stage attenuated malaria parasites. *Sci. Rep.* 6:26824. doi: 10.1038/srep26824
- Labaied, M., Harupa, A., Dumpit, R. F., Coppens, I., Mikolajczak, S. A., and Kappe, S. H. (2007). *Plasmodium yoelii* sporozoites with simultaneous deletion of P52 and P36 are completely attenuated and confer sterile immunity against infection. *Infect. Immun.* 75, 3758–3768. doi: 10.1128/IAI.00225-07
- Lau, L. S., Fernandez-Ruiz, D., Mollard, V., Sturm, A., Neller, M. A., Coziynsen, A., et al. (2014). CD8+ T cells from a novel T cell receptor transgenic mouse induce liver-stage immunity that can be boosted by blood-stage infection in rodent malaria. *PLoS Pathog.* 10:e1004135. doi: 10.1371/journal.ppat.1004135
- Leiriao, P., Mota, M. M., and Rodriguez, A. (2005). Apoptotic Plasmodiuminfected hepatocytes provide antigens to liver dendritic cells. J. Infect. Dis. 191, 1576–1581. doi: 10.1086/429635
- Lindner, S. E., Sartain, M. J., Hayes, K., Harupa, A., Moritz, R. L., Kappe, S. H., et al. (2014). Enzymes involved in plastid-targeted phosphatidic acid synthesis are essential for *Plasmodium yoelii* liver-stage development. *Mol. Microbiol.* 91, 679–693. doi: 10.1111/mmi.12485
- MacKellar, D. C., O'Neill, M. T., Aly, A. S. I., Sacci, J. B. Jr., Cowman, A. F., and Kappe, S. H. I. (2010). *Plasmodium falciparum* PF10-0164 (ETRAMP10.3) is an essential parasitophorous vacuole and exported protein in blood stages. *Eukaryot. Cell* 9, 784–794. doi: 10.1128/EC.00336-09

- Malik, A., Egan, J. E., Houghten, R. A., Sadoff, J. C., and Hoffman, S. L. (1991). Human cytotoxic T lymphocytes against the *Plasmodium falciparum* circumsporozoite protein. *Proc. Natl. Acad. Sci. U.S.A.* 88, 3300–3304. doi: 10.1073/pnas.88.8.3300
- Matuschewski, K. (2013). Murine infection models for malaria vaccine development. The malaria example. *Hum. Vaccin. Immunother.* 9, 450–456. doi: 10.4161/hv.23218
- Matuschewski, K., Hafalla, J. C., Borrmann, S., and Friesen, J. (2011). Arrested *Plasmodium* liver stages as experimental anti-malaria vaccines. *Hum. Vaccin.* 7, 16–21. doi: 10.4161/hv.7.0.14557
- Matuschewski, K., Ross, J., Brown, S., Kaiser, K., Nussenzweig, V., and Kappe, S. H. I. (2002). Infectivity-associated changes in the transcriptional repertoire of the malaria parasite sporozoite stage. *J. Biol. Chem.* 277, 41948–41953. doi: 10.1074/jbc.M207315200
- Mellouk, S., Maheshwari, R. K., Rhodes-Feuillette, A., Beaudoin, R. L., Berbiguier, N., Matile, H., et al. (1987). Inhibitory activity of interferons and interleukin 1 on the development of *Plasmodium falciparum* in human hepatocyte cultures. *J. Immunol.* 139, 4192–4195.
- Mikolajczak, S. A., Lakshmanan, V., Fishbauger, M., Camargo, N., Harupa, A., Kaushansky, A. L., et al. (2014). A next-generation genetically attenuated *Plasmodium falciparum* parasite created by triple gene deletion. *Mol. Ther.* 22, 1707–1715. doi: 10.1038/mt.2014.85
- Mueller, A. K., Camargo, N., Kaiser, K., Andorfer, C., Frevert, U., Matuschewski, K., et al. (2005b). *Plasmodium* liver stage developmental arrest by depletion of a protein at the parasite-host interface. *Proc. Natl. Acad. Sci. U.S.A.* 102, 3022–3027. doi: 10.1073/pnas.0408442102
- Mueller, A. K., Deckert, M., Heiss, K., Goetz, K., Matuschewski, K., and Schlüter, D. (2007). Genetically attenuated *Plasmodium berghei* liver stages persist and elicit sterile protection primarily via CD8 T cells. *Am. J. Pathol.* 171, 107–115. doi: 10.2353/ajpath.2007.060792
- Mueller, A. K., Labaied, M., Kappe, S., H., and Matuschewski, K. (2005a). Genetically modified *Plasmodium* parasites as a protective experimental malaria vaccine. *Nature* 433, 164–167. doi: 10.1038/nature03188
- Müller, K., Gibbins, M. P., Matuschewski, K., and Hafalla, J. C. (2017). Evidence of cross-stage CD8+ T cell epitopes in malaria pre-erythrocytic and blood stage infections. *Parasite Immunol.* 39:e12434. doi: 10.1111/pim.12434
- Müller, K., Matuschewski, K., and Silvie, O. (2011). The Puf-family RNA-binding protein Puf2 controls sporozoite conversion to liver stages in the malaria parasite. *PLoS ONE* 6:e19860. doi: 10.1371/journal.pone.0019860
- Murphy, S. C., Kas, A., Stone, B. C., and Bevan, M. J. (2013). A T-cell response to a liver-stage *Plasmodium* antigen is not boosted by repeated sporozoite immunizations. *Proc. Natl. Acad. Sci. U.S.A.* 110, 6055–6060. doi: 10.1073/pnas.1303834110
- Nagel, A., Prado, M., Heitmann, A., Tartz, S., Jacobs, T., Deschermeier, C., et al. (2013). A new approach to generate a safe double-attenuated *Plasmodium* liver stage vaccine. *Int. J. Parasitol.* 43, 503–514. doi: 10.1016/j.ijpara.2013.01.010
- Nganou-Makamdop, K., Ploemen, I., Behet, M., van Germert, G. J., Hermsen, C., Roestenberg, M., et al. (2012). Reduced *Plasmodium berghei* sporozoite liver load associates with low protective efficacy after intradermal immunization. *Parasite Immunol.* 34, 562–569. doi: 10.1111/pim.12000.x
- Nussenzweig, R. S., Vanderberg, J., Most, H., and Orton, C. (1967). Protective immunity produced by the injection of X-irradiated sporozoites of *Plasmodium berghei. Nature* 216, 160–162. doi: 10.1038/216160a0
- Nussenzweig, R. S., Vanderberg, J. P., Most, H., and Orton, C. (1969). Specificity of protective immunity produced by X-irradiated *Plasmodium berghei* sporozoites. *Nature* 222, 488–489. doi: 10.1038/222488a0
- Offeddu, V., Thathy, V., Marsh, K., and Matuschewski, K. (2012). Naturally acquired immune responses against *Plasmodium falciparum* sporozoites and liver infection. *Int. J. Parasitol.* 42, 535–548. doi: 10.1016/j.ijpara.2012.03.011
- Olotu, A., Fegan, G., Wambua, J., Nyangweso, G., Leach, A., Lievens, M., et al. (2016). Seven-year efficacy of RTS,S/AS01 malaria vaccine among young African children. *New Engl. J. Med.* 374, 2519–2529. doi: 10.1056/NEJMoa1515257
- Orito, Y., Ishino, T., Iwanaga, S., Kaneko, I., Kato, T., Ménard, R., et al. (2013). Liver-specific protein 2: a *Plasmodium* protein exported to the hepatocyte cytoplasm and required for merozoite formation. *Mol. Microbiol.* 87, 66–79. doi: 10.1111/mmi.12083

- Pei, Y., Tarun, A. S., Vaughan, A. M., Herman, R. W., Soliman, J. M., Erickson-Wayman, A., et al. (2010). *Plasmodium* pyruvate dehydrogenase activity is only essential for the parasite's progression from liver infection to blood infection. *Mol. Microbiol.* 75, 957–971. doi: 10.1111/j.1365-2958.2009.07034.x
- Ploemen, I. H., Croes, H. J., van Germert, G. J., Wijers-Rouw, M., Mernsen, C. C., and Sauerwein, R. W. (2012). *Plasmodium berghei* ∆p52&p36 parasites develop independent of a parasitophorous vacuole membrane in Huh-7 liver cells. *PLoS ONE* 7:e50772. doi: 10.1371/journal.pone.0050772
- Prudêncio, M., Rodriguez, A., and Mota, M. M. (2006). The silent path to thousands of merozoites: the *Plasmodium* liver stage. *Nat. Rev. Microbiol.* 4, 849–856. doi: 10.1038/nrmicro1529
- Putrianti, E. D., Silvie, O., Kordes, M., Borrmann, S., and Matuschewski, K. (2009). Vaccine-like immunity against malaria by repeated causal-prophylactic treatment of liver-stage *Plasmodium* parasites. J. Infect. Dis. 199, 899–903. doi: 10.1086/597121
- Renia, L., Maranon, C., Hosmalin, A., Gruner, A. C., Silvie, O., and Snounou, G. (2006). Do apoptotic *Plasmodium*-infected hepatocytes initiate protective immune responses? *J. Infect. Dis.* 193, 163–164. doi: 10.1086/498536
- Rizopoulos, Z., Matuschewski, K., and Haussig, J. (2016). Distinct prominent roles of enzymes of *Plasmodium berghei* heme biosynthesis in sporozoite and liver stage maturation. *Infect. Immun.* 84, 3252–3262. doi: 10.1128/IAI.00148-16
- Rodrigues, M., Nussenzweig, R. S., and Zavala, F. (1993). The relative contribution of antibodies, CD4+ and CD8+ T cells to sporozoite-induced protection against malaria. *Immunol.* 80, 1–5.
- Roestenberg, M., McCall, M., Hopman, J., Wiersma, J., Luty, A. J., van Gemert, G. J., et al. (2009). Protection against a malaria challenge by sporozoite inoculation. *New Engl. J. Med.* 361, 468–477. doi: 10.1056/NEJMoa0805832
- Romero, P., Maryanski, J. L., Corradin, G., Nussenzweig, R. S., Nussenzweig, V., et al. (1989). Cloned cytotoxic T cells recognize an epitope in the circumsporozoite protein and protect against malaria. *Nature* 341, 323–326. doi: 10.1038/341323a0
- RTS,S Clinical Trials Partnership (2011). First results of phase 3 trial of RTS, S/AS01 malaria vaccine in African children. New Engl. J. Med. 365, 1863–1875. doi: 10.1056/NEJMoa1102287
- Sano, G., Hafalla, J. C., Morrot, A., Abe, R., Lafaille, J. J., and Zavala, F. (2001). Swift development of protective effector functions in naive CD8(+) T cells against malaria liver stages. J. Exp. Med. 194, 173–180. doi: 10.1084/jem.194.2.173
- Scheller, L. F., and Azad, A. F. (1995). Maintenance of protective immunity against malaria by persistent hepatic parasites derived from irradiated sporozoites. *Proc. Natl. Acad. Sci. U.S.A.* 92, 4066–4068. doi: 10.1073/pnas.92.9.4066
- Scheller, L. F., Wirtz, R. A., and Azad, A. F. (1994). Susceptibility of different strains of mice to hepatic infection with *Plasmodium berghei*. *Infect. Immun.* 62, 4844–4847.
- Schofield, L., Villaquiran, J., Ferreira, A., Schellekens, H., Nussenzweig, R., et al. (1987). Gamma interferon, CD8+ T cells and antibodies required for immunity to malaria sporozoites. *Nature* 330, 664–666. doi: 10.1038/330664a0
- Shears, M. J., MacRae, J. I., Mollard, V., Goodman, C. D., Sturm, A., Orchard, L. M., et al. (2017). Characterization of the *Plasmodium falciparum* and P. berghei glycerol 3-phosphate acytransferase involved in FASII fatty acid utilization in the malaria parasite apicoplast. *Cell. Microbiol.* 19:e12633. doi: 10.1111/cmi.12633
- Shekalaghe, S., Rutaihwa, M., Billingsley, P. F., Chemba, M., Daubenberger, C. A., James, E. R., et al. (2014). Controlled human malaria infection of Tanzanians by intradermal injection of aseptic, purified, cryopreserved *Plasmodium falciparum* sporozoites. *Am. J. Trop. Med. Hyg.* 91, 471–480. doi: 10.4269/ajtmh.14-0119
- Silvie, O., Goetz, K., and Matuschewski, K. (2008b). A sporozoite asparaginerich protein controls initiation of *Plasmodium* liver stage development. *PLoS Pathog.* 4:e1000086. doi: 10.1371/journal.ppat.1000086
- Silvie, O., Matuschewski, K., Mota, M. M., and Prudêncio, M. (2008a). Interactions of the malaria parasite and its mammalian host. *Curr. Opin. Microbiol.* 11, 352–359. doi: 10.1016/j.mib.2008.06.005
- Silvie, O., Semblat, P., Franetich, J. F., Hannoun, L., Eling, W., and Mazier, D. (2002). Effects of irradiation on *Plasmodium falciparum* sporozoite hepatic development: implications for the design of pre-erythrocytic malaria vaccines. *Parasite Immunol.* 24, 221–223. doi: 10.1046/j.1365-3024.2002. 00450.x

- Spielmann, T., Montagna, G. N., Hecht, L., and Matuschewski, K. (2012). Molecular make-up of the *Plasmodium* parasitophorous vacuolar membrane. *Int. J. Med. Microbiol.* 302, 179–186. doi: 10.1016/j.ijmm.2012.07.011
- Spring, M., Murphy, J., Nielsen, R., Dowler, M., Bennett, J. W., Zarling, S., et al. (2013). First-in-human evaluation of genetically attenuated *Plasmodium falciparum* sporozoites administered by bite of *Anopheles* mosquitoes to adult volunteers. *Vaccine* 31, 4975–4983. doi: 10.1016/j.vaccine.2013.08.007
- Stoute, J. A., Slaoui, M., Heppner, D. G., Momin, P., Kester, K. E., Desmons, P., et al. (1997). A preliminary evaluation of a recombinant circumsporozoite protein vaccine against *Plasmodium falciparum* malaria. *New Engl. J. Med.* 336, 86–91. doi: 10.1056/NEJM199701093360202
- Tarun, A. S., Dumpit, R. F., Camargo, N., Labaied, M., Liu, P., Takagi, A., et al. (2007). Protracted sterile protection with *Plasmodium yoelii* pre-erythrocytic genetically attenuated parasite malaria vaccines is independent of significant liver-stage persistence and is mediated by CD8+ T cells. *J. Infect. Dis.* 196, 608–616. doi: 10.1086/519742
- Trimnell, A., Takagi, A., Gupta, M., Richie, T. L., Kappe, S. H., and Wang, R. (2009). Genetically attenuated parasite vaccines induce contact-dependent CD8+ T cell killing of *Plasmodium yoelii* liver stage-infected hepatocytes. *J. Immunol.* 183, 5870–5878. doi: 10.4049/jimmunol.0900302
- van Buskirk, K. M., O'Neill, M. T., de la Vega, P., Maier, A. G., Krzych, U., Williams, J., et al. (2009). Preerythrocytic live-attenuated *Plasmodium falciparum* vaccine by design. *Proc. Natl. Acad. Sci. U.S.A.* 106, 13004–13009. doi: 10.1073/pnas.0906387106
- van Dijk, M. R., Douradinha, B., Franke-Fayard, B., Heussler, V., van Dooren, M. W., van Schaijk, B., et al. (2005). Genetically attenuated, P36p-deficient malarial sporozoites induce protective immunity and apoptosis of infected liver cells. *Proc. Natl. Acad. Sci. U.S.A.* 102, 12194–12199. doi: 10.1073/pnas.0500925102
- van Schaijk, B. C., Janse, C. J., van Gemert, G. J., van Dijk, M. R., Gego, A., Franetich, J. F., et al. (2008). Gene disruption of *Plasmodium falciparum* p52 results in attenuation of malaria liver stage development in cultured primary human hepatocytes. *PLoS ONE* 3:e3549. doi: 10.1371/journal.pone.0003549
- van Schaijk, B. C., Kumar, T. R., Vos, M. W., Richman, A., van Germert, G. J., Li, T., et al. (2014b). Type II fatty acid biosynthesis is essential for *Plasmodium falciparum* sporozoite development in the midgut of *Anopheles* mosquitoes. *Eukaryot. Cell* 13, 550–559. doi: 10.1128/EC.00264-13
- van Schaijk, B. C., Ploemen, I. H., Annoura, T., Vos, M. W., Foquet, L., van Gemert, G. J., et al. (2014a). A genetically attenuated malaria vaccine candidate based on *P. falciparum* b9/slarp gene-deficient sporozoites. *Elife* 3:e03582. doi: 10.7554/eLife.03582
- Vaughan, A. M., O'Neill, M. T., Tarun, A. S., Camargo, N., Phuong, T. M., Aly, A. S. I., et al. (2009). Type II fatty acid synthesis is essential only for

malaria parasite late liver stage development. Cell. Microbiol. 11, 506-520. doi: 10.1111/j.1462-5822.2008.01270.x

- Weiss, W. R., Sedegah, M., Beaudoin, R. L., Miller, L. H., and Good, M. F. (1988). CD8+ T cells (cytotoxic/suppressors) are required for protection in mice immunized with malaria sporozoites. *Proc. Natl. Acad. Sci. U.S.A.* 85, 573–576. doi: 10.1073/pnas.85.2.573
- White, K. L., Snyder, H. L., and Krzych, U. (1996). MHC class I-dependent presentation of exoerythrocytic antigens to CD8+ T lymphocytes is required for protective immunity against *Plasmodium berghei*. J. Immunol. 156, 3374–3381.
- WHO (2016a). World Malaria Report 2016. Geneva: World Health Organisation.
- WHO (2016b). Tables of Malaria Vaccine Projects Globally: The Rainbow Table. Geneva: World Health Organization.
- Yu, M., Santah Kumar, T. R., Nkroumah, L. J., Coppi, A., Retzlaff, S., Li, C. D., et al. (2008). The fatty acid biosynthesis enzyme FabI plays a key role in the development of liver-stage malarial parasites. *Cell Host Microbe* 4, 567–578. doi: 10.1016/j.chom.2008.11.001
- Zhang, M., Fennell, C., Ranford-Cartwright, L., Sakthivel, R., Gueirard, P., Meister, S., et al. (2010). The *Plasmodium* eukaryotic initiation factor-2alpha kinase IK2 controls the latency of sporozoites in the mosquito salivary glands. *J. Exp. Med.* 207, 1465–1474. doi: 10.1084/jem.20091975
- Zhang, M., Mishra, S., Sakthivel, R., Fontoura, B. M., and Nussenzweig, V. (2016). UIS2: a unique phosphatase required for the development of *Plasmodium* liver stages. *PLoS Pathog.* 12:e1005370. doi: 10.1371/journal.ppat.10 05370

Conflict of Interest Statement: KMa is listed as inventor on international and national patents "Live genetically attenuated malaria vaccine" and "Live genetically engineered protozoan vaccine." These patents were filed by the authors' non-profit institutions to promote the development and distribution of malaria vaccines to people in need worldwide, in accordance with a global access strategy.

The other authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Copyright © 2017 Kreutzfeld, Müller and Matuschewski. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) or licensor are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.